Author's response to reviews

Title: Role of the splicing factor SRSF4 in cisplatin-induced modifications of pre-mRNA splicing and apoptosis.

Authors:

Maude Gabriel (maude.gabriel@ulg.ac.be)
Yves Delforge (yves.delforge@ulg.ac.be)
Adeline Deward (adeward@ulg.ac.be)
Yvette Habraken (Yvette.habraken@ulg.ac.be)
Benoit Hennuy (benoit.hennuy@ulg.ac.be)
Jacques Piette (jpiette@ulg.ac.be)
Roscoe Klinck (roscoe.klinck@usherbrooke.ca)
Benoit Chabot (benoit.chabot@usherbrooke.ca)
Alain Colige (acolige@ulg.ac.be)
Charles Lambert (c.lambert@ulg.ac.be)

Version: 2
Date: 19 November 2014

Author's response to reviews:

Dear Editor,

We are pleased to resubmit the manuscript (ID1653816502149879) entitled “Role of the splicing factor SRSF4 in cisplatin-induced modifications of pre-mRNA splicing and apoptosis.” to be considered for publication in BMC cancer as an original research article.

We performed changes as requested. We included line numbers and the approval of the ethic committee of Liège University Hospital in the Methods section (line 109).

We thank you for your consideration and are looking forward to the editorial decision.

Sincerely yours,

M. Gabriel, MSc, corresponding author
A. Colige, Professor, promoter of the work
C. Lambert, PhD, promoter of the work